Author's response to reviews

Title: Development of the adult and child complementary medicine questionnaires fielded on the National Health Interview Survey.

Authors:

Barbara J Stussman (stussmanbi@mail.nih.gov)
Christina D Bethell (bethellc@ohsu.edu)
Caroline Gray (cpgray@gmail.com)
Richard L Nahin (nahinr@mail.nih.gov)

Version: 4 Date: 15 November 2013

Author's response to reviews: see over
Development of the adult and child complementary medicine questionnaires fielded on the National Health Interview Survey.
Barbara J Stussman, Christina D Bethell, Caroline Gray and Richard L Nahin

To the Editors of BMC CAM:

We thank the Executive Editor for his comments concerning our manuscript under review. We have incorporated these as indicated below.

Editorial comments:

Please confirm whether informed consent was obtained from participants in the interviews that were performed as part of your study, and include a statement to this effect in your manuscript.

Response to Reviewer:

We have now indicated that informed consent was obtained from participants in cognitive interviews and focus groups. On the bottom of page 15 we say, “Informed consent was obtained from all participants prior to the start of the interview.” In the 2nd paragraph on page 22 we say, “Informed consent was obtained from all interview and focus group participants”.

We hope these final revisions result in a prompt publication.